Author's response to reviews

Title: When is perigraft air pathological?

Authors:

Elizabeth Ball (liz_ball@yahoo.com)
Gareth Morris-Stiff (garethmorrisstiff@hotmail.com)
Mari Coxon (maricoxon@hotmail.com)
Michael H Lewis (mike.lewis@pr-tr.wales.nhs.uk)

Version: 3 Date: 11 June 2007

Author's response to reviews: see over
When is perigraft air pathological?

Replies to reviewers comments

Reviewer 1 – Samuel Wong

1. All the small grammatical errors have been addressed.

2. Early postoperative phase has been defined.

3. Discussion has been expanded to include potential causes for the air.

Reviewer 2 – Wing Siu

1. A sentence on supplementary tests has been included.

2. More details on the patient’s symptoms have been added.

3. Sentence on haemostatic agents has been added.

4. Sentence added on type of CT scan performed.

5. Discussion on other relevant findings has been added.